Author’s response to reviews

Title: Osteoarticular lesion in xanthoma disseminatum treated with total hip arthroplasty: a case report

Authors:

Yu Taniguchi (cindy.forever911@gmail.com)
Tomofumi Nishino (nishino@md.tsukuba.ac.jp)
Hisashi Sugaya (h.sugaya.1205@tsukuba-seikei.jp)
Hajime Mishima (hmishima@md.tsukuba.ac.jp)
Naoyuki Ochiai (ochiainaoyuki@ob.md.tsukuba.ac.jp)
Masashi Yamazaki (masashiy@md.tsukuba.ac.jp)

Version: 1 Date: 29 Jan 2019

Author’s response to reviews:

Chinmoy Bose, M.D., Ph.D.
Editor-in-Chief, “Journal of Medical Case Reports”

Yu Taniguchi, M.D., Ph.D.
Division of Regenerative Medicine for Musculoskeletal System,
Department of Orthopaedics Surgery, Faculty of Medicine,
University of Tsukuba,
1-1-1 Tennodai,
Tsukuba, Ibaraki, 305-8575
Japan
E-mail: cindy.forever911@tsukuba-seikei.jp
31th January, 2019
Dear Chinmoy Bose, M.D., Ph.D.

We would like to thank you for giving us the opportunity to revise our manuscript. We have carefully read all the remarks made by the reviewer and addressed all of the comments. All the passages that we have changed or added to the manuscript are highlighted in gray.

Once again, thank you for considering our manuscript.

Sincerely yours,

Yu Taniguchi, M.D., Ph.D.

Major points of correction in the revised manuscript are as follows:

Response to Reviewer

#1 Reviewer

5. Is the interpretation (discussion and conclusion) well balanced and supported by the case presented?

Comments: yes, maybe conclusion is too short, especially regarding follow up

#4 Reviewer

The authors are thanked for their submission.

However, we'd ask them to shorten the case presentation slightly and expand the discussion.

Answer)

Thank you for your comment. In order to shorten the case presentation, I deleted the sentences in the elbow and knee joints that have nothing to do with the progress of the hip joint. (P5 Line 106-108)
Regarding discussion, since this case is a very rare case, it is difficult to expand the discussion part, so we did not add or modify sentences.